Author's response to reviews

Title: Child sexual abuse in southern Brazil and associated factors: a population-based study

Authors:

Diego G Bassani (bassanid@smh.toronto.on.ca)
Lilian S Palazzo (lspalazzo@hotmail.com)
Jorge U Beria (j.u.b@terra.com.br)
Luciana P Gigante (luciana@homemail.com)
Andreia CL Figueiredo (andreialfig@terra.com.br)
Denise RGC Aerts (daerts@via-rs.net)
Beatriz CW Raymann (bearaymann@terra.com.br)

Version: 5 Date: 29 April 2009

Author's response to reviews:

Dear editors,

The corrected version of the manuscript proof has been uploaded to BMC's system. Please contact me if you require further changes.

Regards,

Diego Bassani

Diego G. Bassani, M.Sc., Ph.D.
Epidemiologist - Maternal & Child Health
Centre for Global Health Research - Li Ka Shing Knowledge Institute, St Michael's Hospital
Assistant Professor - Dalla Lana School of Public Health - Faculty of Medicine, University of Toronto
Adjunct Scientist - Child Health and Evaluative Sciences - Hospital for Sick Children

70 Richmond Street East, Suite 202A - Toronto, ON - M5C 1N8, Canada
Tel: +1 416 864 6060 ext. 3357 | Fax: +1 416 964 5256
www.cghr.org | www.phs.utoronto.ca | www.sickkids.ca/respgreatnet